Author's response to reviews

Title: Gender differences in mortality and risk factors in a 13-years cohort study of street-recruited injecting drug users

Authors:

Linn Gjersing (lgjersing@gmail.com)
Anne Line Bretteville-Jensen (alb@sirus.no)

Version: 3
Date: 25 April 2014

Author's response to reviews: see over
Cover letter

To the Editor at BMC Public Health

Thank you for your interest in our paper and for allowing us to revise and resubmit the manuscript.

We have edited the paper by replacing the term “prostitution” with the term “sex work” as requested by the Editor. We have also had the paper proofread to ensure that it adhere to BMC Public Health standards. We hope the revised paper will be acceptable according to the Journal’s standards.

Please do not hesitate to contact us again should there be further issues that need to be addressed.

Kind Regards,
Linn Gjersing Ph.D
Researcher

SIRUS-Norwegian Institute for Alcohol and Drug Research
Øvre Slottsgate 2B, Oslo
P.O.Box 565 Sentrum,
NO-0105 Oslo, Norway
Ph +47 40 64 87 11 / Mob +47 934 10 236
Email: lg@sirus.no